Author's response to reviews

Title: Managing change in the nursing shift handover from traditional to bedside handover - a case study from Mauritius

Authors:

Hemant K Kassean (h.kassean@uom.ac.mu)
Zaheeda B Jagoo (zjagoo@yahoo.com)

Version: 5 Date: 7 December 2004

Author's response to reviews:

All changes have been made as advised, except for point 4. On p.3 par 1 add the word nurses between the words 'about' and 'not being involved' - this sentence has been rephrased to make the meaning clearer.